CS 415 � Practicum in Operating Systems

Student Information Sheet

(Please fill and return this form to the instructor at the end of class)

• Name (Last, First):

• CUNetId:

• CUID:

• Current Status (Undergrad/grad, which year):

• Major Field:

• You are taking this course for:

o Letter
o Audit
o S/U
o Other

• Please describe in a few words the past experience you have had in working with computer systems. It includes the courses you have taken and any other experience you may have had, like in the industry or home.

• Which Operating Systems have you worked with before, and to what extent?

o Unix:
o Windows:
o Linux:
o DOS:
o Others (please specify):

• Which languages are you well versed in?

o C
o C++
o Java
o Other